Author's response to reviews

Title: Quadricuspid aortic valve not discovered by transthoracic echocardiography

Authors:

Magnus Dencker (magnus.dencker@skane.se)
Martin Stagmo (martin.stagmo@skane.se)

Version: 2 Date: 26 October 2006

Author's response to reviews: see over
Dear Dr Eugenio Picano
Editor-in-Chief
Cardiovascular Ultrasound

We have altered the revised manuscript according to the suggestions by the reviewer.

Reviewer 1’s report
An interesting and well presented case report describing a quadricuspid aortic valve, type E.
Please, correct minor spelling mistakes.
1) page 2: incrasing should be increasing

Comment: This has been corrected.
2) the same on page 3

Comment: This has been corrected.
3) page 4: transthorasic should be transthoracic.

Comment: This has been corrected.

Technical revision
Requested formatting changes
Title Page/Page 1
Please remove all unnecessary capitalisation from the title

Comment: This has been corrected.
Please remove author’s qualifications.

Comment: This has been corrected.
Please correct to Corresponding author.

Comment: This has been corrected.
Please provide all authors’ e-mail addresses.

Comment: This has been corrected.
Please provide Keywords after the corresponding author.

Comment: This has been corrected.

Page 2
Line 7: Please correct to “increasing dyspnea”.

Comment: This has been corrected and now read; “increasing dyspnoea”

Page 3
Line 4: Please correct to [1,2,3,4,5,6,7,8]. References must be numbered consecutively.

Comment: This has been corrected.
Line 6: Please correct to “increasing dyspnea”.

Comment:This has been corrected and now read; “increasing dyspnoea”
Lines 13,18: Please correct to “(Figure 1 in diastole and Figure 2 in systole)”, and “(Figure 3).

Comment: This has been corrected.

Page 4
Line 13: Please correct to “transthoracic versus…”.

Comment: This has been corrected.
Line 18: Please correct to Conclusions.

Comment: This has been corrected.

Page 5
Please correct to “The authors declare…”.
Comment: This has been corrected.
Page 7
First line: Please correct to Figure legends (no semicolon).
Comment: This has been corrected.
Lines 2,3,4: Please correct to Figure 1, Figure 2, and Figure 3.
Comment: This has been corrected.